Author's response to reviews

Title: A population-based survey on tanning bed use in Germany

Authors:

Franziska U Börner (f.boerner@fz-juelich.de)
Holger Schütz (h.schuetz@fz-juelich.de)
Peter Wiedemann (p.wiedemann@fz-juelich.de)

Version: 6 Date: 14 July 2009

Author's response to reviews:

Dear Editor,

we have made the final revisions and requested corrections in our manuscript. We have added the online source for reference 26, deleted the figure title and cropped the image, removed line numbers, keywords, vertical table lines and so on. We also took the opportunity to check our manuscript again and hope that everything will no be to your satisfaction.

Sincerely

Franziska Börner